@AGU PUBLICATIONS

Geophysical Research Letters

RESEARCH LETTER

Kev Points:

- Net carbon sink represented by net biome productivity (NBP) has decreased for terrestrial Indian region in the last three decades
- · Land cover change and climate change drive the decrease in NBP from 1980s to 1990s and 1990s to 2000s, respectively
- · Rise in average temperatures has led to significant loses of carbon from ecosystems in the form of respiration

Supporting Information:

Supporting Information S1

Correspondence to:

A. K. Jain, & S. Baidya Roy, iain1@illinois.edu: drsbr@iitd.ac.in

Citation:

Gahlot, S., Shu, S., Jain, A. K., & Baidya Roy, S. (2017). Estimating trends and variation of net biome productivity in India for 1980-2012 using a land surface model. Geophysical Research Letters, 44. https://doi.org/10.1002/ 2017GL075777

Received 16 MAR 2017 Accepted 31 OCT 2017 Accepted article online 6 NOV 2017

10.1002/2017GL075777

Estimating Trends and Variation of Net Biome Productivity in India for 1980-2012 Using a Land Surface Model

Shilpa Gahlot¹, Shijie Shu², Atul K. Jain², and Somnath Baidya Roy¹

¹Center for Atmospheric Sciences, Indian Institute of Technology Delhi, New Delhi, India, ²Department of Atmospheric Sciences, University of Illinois, Urbana, IL, United States

Abstract In this paper we explore the trend in net biome productivity (NBP) over India for the period 1980–2012 and quantify the impact of different environmental factors, including atmospheric CO_2 concentrations ([CO₂]), land use and land cover change, climate, and nitrogen deposition on carbon fluxes using a land surface model, Integrated Science Assessment Model. Results show that terrestrial ecosystems of India have been a carbon sink for this period. Driven by a strong CO₂ fertilization effect, magnitude of NBP increased from 27.17 TgC/yr in the 1980s to 34.39 TgC/yr in the 1990s but decreased to 23.70 TgC/yr in the 2000s due to change in climate. Adoption of forest conservation, management, and reforestation policies in the past decade has promoted carbon sequestration in the ecosystems, but this effect has been offset by loss of carbon from ecosystems due to rising temperatures and decrease in precipitation.

Plain Language Summary Our study presents the results for net biome productivity (NBP) for India on a country scale, which are estimated using a state-of-the-art land surface model. Environmental factors, such as elevated atmospheric CO₂, climate change, nitrogen deposition, and land cover changes, have been identified as the major factors responsible for changes in terrestrial carbon fluxes. Broadly, the manuscript presents the relative contribution of these environmental factors to changes in NBP in India for the last three decades. Most of the published studies on this topic have focused on the NBP estimates at regional or global scale. To the best of our knowledge, this is the first attempt to quantify NBP at country scale. The results presented are timely because this manuscript adds to the knowledge of the carbon budget on a country level, identifies uncertainties in our understanding that can motivate further experiments, and aides in guantifying the regional and global carbon budget. These original research results presented in this study have implications for the ongoing international debate on climate treaty which calls for countries to apply the carbon stored in its forests and other ecosystems toward its budgeted reduction in carbon dioxide and other greenhouse gases.

1. Introduction

Quantification of the exchange of carbon between the atmosphere and terrestrial biosphere is central to monitoring the influence of anthropogenic disturbances on the terrestrial ecosystems. This exchange is well represented by the net biome productivity (NBP = $GPP - Ra - Rh - E_{luc}$) which considers gross carbon assimilation or gross primary productivity (GPP), total ecosystem respiration including both autotrophic (Ra) and heterotrophic (Rh) respirations, and anthropogenic carbon fluxes, such as net emissions due to land use and land cover change (E_luc) for a given ecosystem and for a region or globe. The ecosystem acts as a net carbon source or carbon sink, if the NBP flux values are negative or positive, respectively. There are numerous global-scale carbon flux studies (Beer et al., 2010; Le Quéré et al., 2015), but very few have focused on India. Most of the studies on India (Bala et al., 2013; Banger et al., 2015; Goroshi et al., 2014; Nayak et al., 2010) estimate the net primary productivity (NPP) by vegetation, (NPP = GPP - Ra) and do not account for soil carbon fluxes (e.g., soil heterotrophic respiration Rh), and anthropogenic disturbances, for example, E_luc. Majority of the studies that calculate NPP have used empirical models, like Carnegie-Ames-Stanford Approach (CASA) (Dadhwal et al., 2011; Goroshi et al., 2014; Nayak et al., 2015) and light use efficiency model implemented for the MODIS sensor (Bala et al., 2013). Only Banger et al. (2015) have used a process-based model, Dynamic Land Ecosystem Model, to study the role of environmental factors on NPP in India. Empirical models typically need a wide range of remote sensing data and hence are confined to the recent decades. In contrast, process-based models can be used to study the presatellite era as well as the future.

©2017. American Geophysical Union. All Rights Reserved.

More importantly, process-based models can be used to conduct numerical experiments to derive insights into the processes and interactive feedback mechanisms that are responsible for variation in carbon fluxes with time.

Nayak et al. (2015) and Nayak et al. (2016) have studied net ecosystem production (NEP = NPP – Rh) that does not account for anthropogenic flux. Using CASA model, Nayak et al. (2015) found India to be a net carbon sink with a total annual averaged uptake of 9.9 TgC/yr over the time period 1981–2006. Nayak et al. (2016) reported 42 TgC/yr and 18 TgC/yr as average NEP values for India from 2001 to 2006 from MODIS-based and AVHRR-based estimates, respectively. By taking ensemble average of many approaches, Cervarich et al. (2016) found NEP and NBP values for India to be in the 200.6 \pm 137.7 TgC/yr and 185.9 \pm 145.6 TgC/ yr range, respectively, for the 2000–2013 period. It is evident from this literature review that information on long-term variability and drivers of NBP over India is lacking, thereby limiting our understanding of the dynamics of the terrestrial carbon cycle, which is overall objective of the study.

In specific, this study has two main objectives: (1) to estimate the trend of NBP for the last three decades (1980–2012) and (2) to quantify the influence of major environmental and anthropogenic factors on carbon fluxes. We accomplish these objectives using a process-based land surface model, Integrated Science Assessment Model (ISAM), that is driven by forcings of [CO₂], land use and land cover change (LULCC), climate change, and nitrogen deposition to simulate terrestrial carbon dynamics for India from 1801 to 2012.

2. Methods

2.1. Model Description

The carbon fluxes for this study are generated using ISAM that is a state-of-the-art land surface model that simulates carbon, water and energy fluxes at $0.5^{\circ} \times 0.5^{\circ}$ spatial resolution, and multiple temporal resolutions ranging from half hour to yearly time steps. ISAM simulates carbon fluxes through the processes of photosynthesis, carbon allocation to different plant parts, and autotrophic and heterotrophic respirations. Effects of different environmental factors like [CO₂], climate (e.g., temperature and precipitation), and photosynthetically active radiation are considered in flux estimation. Availability and effect of nitrogen is taken into account by fully coupling the carbon and nitrogen cycles in the model. The model accounts for dynamic phenology, root distribution, and depth parameterizations for different ecosystems. Stress factors of light, water, and nutrient limitation are considered while allocating the assimilated carbon to different plant parts (El-Masri et al., 2015; Song et al., 2016, 2013). The structure, parameterization and performance of ISAM have been extensively tested and verified in various studies (Barman et al., 2014a, 2014b; El-Masri et al., 2013, 2015; Jain et al., 2009, 2013).

2.2. Input Data

The ISAM model requires [CO₂], climate, nitrogen deposition, and LULCC data as input. We use [CO₂] data from Global Carbon Project Budget 2015 (Le Quéré et al., 2015). Gridded estimates of airborne nitrogen deposition are from Lamarque et al. (2011). The source of climate data is Climate Research Unit-National Centers for Environmental Prediction reanalysis (Harris et al., 2014). LULCC data (Jain et al., 2013) used have been reconstructed from Historical Database of the Global Environment (HYDE 3.1) (Klein et al., 2011) for cropland and pastureland transitions (Figure S1 in the supporting information). HYDE land cover data set used for driving the model has been compared with land cover data set based on updated version of Ramankutty and Foley (1999) (hereafter referred as SAGE) for India in Table S1. HYDE and SAGE data sets show slightly different levels of cropland and forest cover, the two main vegetation types for India. HYDE data estimated cropland areas over the last three decades are consistently higher than SAGE data, whereas HYDE-based estimated forest areas is lower. As discussed in Meiyappan and Jain (2012), the differences between the cropland and forest areas are a result of adoption of different methods and agricultural inventory data sets by both global landcover data sets. Where HYDE inventory data is based on FAO (2008), SAGE estimates have relied more on national-level census statistics and FAO estimates for recent years.

2.3. Experiment Design and Analysis

As a first step to start the model simulations, we spin-up ISAM to allow the carbon and nitrogen pools to reach a steady state at approximately 1800 levels. $[CO_2]$, LULCC, and nitrogen deposition data are fixed at 1800 levels for the spin-up. Since climate data are available only from 1900 onward, climate data from years

Table 1

Description of Model Simulations Conducted With ISAM for India From 1801 to 2012

	Model simulation				
Environment factor	S _{CON}	S _{CO2}	S _{LUC}	S _{CLI}	S_{N_DEP}
Variable CO ₂ Variable LCLUC Variable climate Variable N deposition	~ ~ ~ ~	X	√ X √	√ √ X √	√ √ √ ×

Note. Tick mark (\checkmark) indicates the environmental factor was varied with time. Cross mark (\varkappa) indicates the factor was held static at initial (assumed zero for nitrogen deposition and LCLUC) value. S_{CON}, S_{CO2}, S_{LUC}, S_{CLI} and S_{N_DEP} are model simulations with respective input environmental factors. For example, S_{CON} is the model simulation where all environment factors were accounted for.

1900 to 1930 are recycled to represent stable background climate. The details of this spin-up process are explained in El-Masri et al. (2013, 2015). Next, we conduct a control simulation (S_{CON}) by running ISAM from 1801 to 2012 with all four environmental and anthropogenic forcings, [CO₂], climate, LULCC, and nitrogen deposition, varying with time. Four additional simulations are conducted by turning off one of the four forcings at a time (Table 1). For the S_{CLI} run, [CO₂], LULCC, and nitrogen deposition are assumed to be the same as for the S_{CON} and the climate data are allowed to vary as used for spin-up experiment. The results of each additional simulation are compared with the S_{CON} simulation to estimate the impact of each forcing on the carbon fluxes. Since NEP does not account for LULCC effect, the NBP for S_{LUC} , which is without land use change, from the NBP for S_{CON} , which includes land use change. The temporal variability of the forcings (Table S2 and Figure S1) and their impacts

on the carbon fluxes (Table 2) are averaged over three approximate decades, 1980s (1980–1989), 1990s (1990–1999), and 2000s (2000–2012), for further analysis.

To test the performance of the model specific for India region, we compare the simulated GPP with the spatially explicit FLUXNET-Multi-Tree Ensemble (MTE) data derived from empirical upscaling of eddy covariance measurements (Jung et al., 2009) and the MOD17A2 GPP data products from MODIS observations (Zhao & Running, 2010). For comparing the satellite-based observation data with ISAM model results both of these observed data sets are spatially aggregated to match the $0.5^{\circ} \times 0.5^{\circ}$ resolution of ISAM. In addition, modelestimated NPP, NEP, and NBP have been compared and validated with other studies following multiple approaches for different study periods.

3. Results

3.1. Model Evaluation

The magnitude and overall spatial patterns of ISAM-simulated GPP broadly match the observed values from the FLUXNET-MTE and MODIS data sets (Figure S2). ISAM is able to capture important features such as the low productive Himalayan tundra ecosystem in the north, arid ecosystems in the northwest, and high productive regions like the Himalayan foothills and agricultural lands in the eastern and central parts of India. However, GPP is overestimated in the natural forest ecosystems in the northeast and Western Ghats but is underestimated in the grasslands and croplands in the northwest region as compared to FLUXNET-MTE. Average ISAM estimated GPP for the period 2001–2008 is 2.9 PgC/yr, which lies between MODIS and FLUXNET-MTE estimates of 2.8 PgC/yr and 3.3 PgC/yr, respectively. Simulated NPP (including the effect of LULCC) and NEP at country scale for different time periods have also been compared with other studies in Tables S3a and S3b. Our estimates of NPP and NEP are lower than most studies that have used remote sensing data with LUE models because the measurement of vegetation indices using remote sensing is prone to overestima-

Table 2

Variation in Model Estimated Impact of Environmental Factors, Including $CO_{2^{n}}$ LULCC and Climate, on the Rate of Change of Decadal Average for NBP from 1980s to 2000s

Decadal Transition	Impact of CO ₂	Impact of LCLUC	Impact of Climate
	(TgC/yr ²)	(TgC/yr ²)	(TgC/yr ²)
1980s to 1990s ^a 1990s to 2000s ^a Net Impact (1980s to 2000s) ^a	0.44 0.61 0.52	-1.29 -0.25 -0.77	-0.16 -1.28 -0.72

^a1980s, 1990s and 2000s are the averages for the periods 1980-1989, 1990-1999 and 2000-2012. The ecosystem acts as a net carbon source or carbon sink, if the NBP flux value is negative or positive.

tion caused by lack of downregulating mechanisms due to limiting factors (Nayak et al., 2010). ISAM-simulated NBP for India has also been compared with results from bottom-up and top-down modeling approaches (Figure S3 and Table 3). NBP for India from other modelling studies was calculated by extracting the spatial grids for India from global NBP estimates. The yearly variations (Figure S3) and the ensemble of decadal averages of NBP from eight TRENDY version 4.0 models and five top-down models have been calculated for India to represent interannual variations and decadal average NBP from bottom-up models and top-down (or atmospheric-inversion) models, respectively. It is worth mentioning here that the top-down model estimated results for NBP are only available for the 1996–2012 (Thompson et al., 2016). Prior to 1996 the top-down model, results are not available, because this region is largely missing

Table 3

Comparison of ISAM Estimated Decadal Average NBP (TgC/yr) for India With Bottom-Up and Top-Down Models

DECADE	1980s ^a	1990s ^a	2000s ^a
Bottom-up models ^b	1.04 ± 46.75	34.65 ± 54.16	18.50 ± 48.40
Top-down models ^c	-	45.22 ± 69.53 ^d	42.12 ± 67.40 ⁶
ISAM	27.17	34.39	23.70

^aThe 1980s, 1990s, and 2000s are the averages for the periods 1980–1989, 1990–1999, and 2000–2012. The ecosystem acts as a net carbon source or carbon sink, if the NBP flux value is negative or positive. Based on TRENDY model intercomparison version 4.0 results (Sitch et al., 2015). In this study we used eight bottom-up model results: CLM4.5 (Oleson et al., 2013), JULES (Clark et al., 2011), LPJ (Sitch et al., 2003), LPJ_GUESS(Ahlström et al., 2012), LPX (Stocker et al., 2014), ORCHIDEE (Krinner et al., 2005), VEGAS (Zeng et al., 2005), and VISIT (Ito & Inatomi, 2012). Uncertainties have been calculated as 1σ (standard deviation) and represent differences between estimates from different models. ^CTop-down model results for the period 1996–2012 are based on Thompson et al. (2016). ^dThe averaged value is for the period 1996–1999, which are calculated based on three top-down model results: ACTM (Patra et al., 2011), CCAM (Rayner et al., 2008), and JMA_CDTM (Sasaki et al., 2003). Uncertainties have been calculated as 1σ (standard deviation) and represent differences between estimates from different models. ^eThe results are based on five top-down model results: ACTM (Patra et al., 2011), CCAM (Rayner et al., 2008), GELCA (Ganshin et al., 2011), JMA_CDTM (Sasaki et al., 2003), and Carbon Tracker-Europe (Peters et al., 2007). Uncertainties have been calculated as 1σ (standard deviation) and represent differences between estimates from different models.

atmospheric CO₂ measurement data prior to 1996 (Patra et al., 2016; Thompson et al., 2016). This comparison shows that ISAM estimates of NBP for India are well within the range of NBP estimates from other approaches and follow the same trend as decadal mean NBP from bottom-up models. However, the estimates of NBP from top-down models for India are many times larger than the estimates from bottom-up models. This large uncertainty in NBP estimates indicates the complexity and sensitivity of the terrestrial ecosystems to changes in environmental factors, which is the major objective of this study. The differences between flux estimates from bottom-up and top-down models can also be attributed to incomplete accounting of all relevant fluxes in each of these modeling approaches. Top-down models have a coarser spatial resolution (> $1.8^{\circ} \times 1.8^{\circ}$) which adds to the large uncertainties in the model estimates. Despite the large differences between top-down and bottom-up model results, it is worth noting that the NBP estimates from top-down models lie within one standard deviation of that from bottom-up models.

3.2. Carbon Fluxes From 1980 to 2012

Over the last three decades, carbon fluxes for India including GPP, Ra, NPP, and Rh have increased at the rates of 6.00 TgC/yr^2 , 4.50 TgC/yr^2 , 1.55 TgC/yr^2 , and 1.85 TgC/yr^2 , respectively (Figure S4). In contrast, NBP does not show any definite trend for the study period. Magnitude of NBP has increased from 27.17 TgC/yr in the 1980s to

34.39 TgC/yr in the 1990s and then decreased to 23.70 TgC/yr in 2000s (Figure 1). The increase in NBP from 1980s to 1990s can be attributed to CO_2 fertilization impact that has led to positive trend for NPP and NEP promoting carbon sequestration potential of terrestrial ecosystems. However, because of higher rate of increase in loses of carbon from the ecosystems in the form of respiration than net carbon sequestered by the plants from 1990s to 2000s, NBP has shown a negative trend for this period. Positive NBP values

50 40 30 20 10 0 1980s 1990s 2000s 50 LULCC CO2 Climate 30 10 -10 -30 1980s 1990s 2000s

Figure 1. Model estimated decadal average (1980s, 1990s, and 2000s) of (top) net biome productivity and (bottom) contribution of different factors to it.

estimated by ISAM from 1980s to 2000s indicate that terrestrial ecosystems of India are a net carbon sink due to enhanced CO_2 fertilization effect, but the reducing trend in NBP of 1.07 TgC/yr² from 1990s to 2000s points to a declining carbon assimilation rate in recent years mainly due to rise in average temperatures and decreased precipitation, resulting an increase in loss of carbon from ecosystems in the form of respiration, and partly due to loss of forest cover in the 1980s and 1990s.

ISAM simulations indicate that majority of changes in carbon fluxes over India are driven by increasing [CO₂], LULCC, and climate change (Figure 1). CO₂ fertilization effect, on one hand, is promoting higher NBP at the rate of 0.52 TgC/yr² (Table 2), but LULCC and climate change contribute to the increasing tendency of the ecosystems to act as carbon sources for the study period 1980s to 2000s at the rates of -0.77 TgC/yr² and -0.72 TgC/yr², respectively (Table 2). However, the overall impact of environmental and anthropogenic factors of [CO₂], LULCC, and climate on carbon fluxes of terrestrial ecosystems of India has not been uniform throughout the last three decades.

 CO_2 fertilization effect is evident in our results as increase in NPP over time with a net positive influence on NEP and NBP. With rise in [CO₂], the CO₂ fertilization effect on NBP is much more pronounced from 1990s to 2000s at 0.60 TgC/yr² than from 1980s to 1990s at 0.44 TgC/yr² (Table 2). Deforestation in natural forestland and conversion of forests to croplands has led carbon flux related to LULCC to act as a major source of carbon for India for most part of the 20th century (Figure S1). LULCC is found to be the most dominant driver of negative NBP trend from 1980s to 1990s, resulting in the carbon loss at the rate of 1.29 TgC/yr^2 (Fig 1). But recent attempts of forest restoration and conservation have led to steady decrease in rates of deforestation and increase in secondary forest cover for India in the last three decades (Figure S1). This has led to reduction in carbon loss from terrestrial ecosystems due to LULCC from 1.29 TgC/yr^2 from 1980s to 1990s to 0.25 TgC/yr^2 from 1990s to 2000s (Table 2). This positive impact of increase in forest cover and reduction in deforestation activites on carbon uptake of ecosystems in recent years has also been mentioned in previous studies (Bala et al., 2013; Banger et al., 2015; Kaul et al., 2009; Nayak et al., 2015, 2016; Patra et al., 2013).

The impact of climate change on carbon cycle remains one of the most uncertain components in carbon studies. Increase in decadal average temperature for the study period is expected to negatively affect plant photosynthetic rates and hence total carbon assimilated by terrestrial Indian region. Warmer temperatures have also led to loss of carbon from the ecosystems due to increase in respiration rates. Decadal average temperature has seen much larger increase from 1990s to 2000s than in the decade prior (Table S2). Decadal average of annual precipitation has been found to increase by 2.44% from 1980s to 1990s but has decreased by 4.77% from 1990s to 2000s, causing a net decrease of 2.45% over the study period. The large increase in temperature and subsequent decrease in precipitation have collectively led to climate change acting as an overall major and most dominant source of declining NBP (Figure 1) from 1990s to 2000s impacting NBP at the rate of -1.28 TgC/yr^2 (Table 2). Such strong influence of climate on carbon uptake of ecosystems calls for our attention and need for more such studies focused on climate change trends and its effects on the terrestrial ecosystems of India specific to different vegetation types and at finer spatial resolution.

The role of nitrogen deposition is found to be negligible for fluxes over India (Figure S5), which indicates that India, being a tropical country, is not nitrogen limited. Hence, results for the case of nitrogen deposition are not discussed in this paper.

NBP in India varies a lot along the spatial expanse of the country (Figure S6). More than 50% of Indian land is covered by croplands (Figure S1) that have a low capacity of carbon sequestration. However, parts of Western Ghats, natural evergreen and deciduous forests in the north-east and regions around the Himalayan foothills are seen to be large carbon sinks with higher NBP.

4. Discussions and Conclusions

This study explores the variability of NBP over India from 1980 to 2012 and how this variability is influenced by environmental and anthropogenic factors like $[CO_2]$, LULCC, climate, and N deposition by conducting numerical experiments with the ISAM model. Results show that terrestrial ecosystems of India have been a carbon sink for the entire study period with positive NBP, but with a negative trend in NBP in the recent decades (1990s to 2000s), suggesting that carbon sequestration potential of the ecosystems is decreasing overtime. Even though CO_2 fertilization continues to increase carbon sequestration in the terrestrial ecosystems, this increase is offset majorly by climate change and to some extent by deforestation activities. Of all the factors studied as a part of this research, climate change is the most uncertain and significant factor that is likely to impact carbon fluxes the most in future, because under climate change the terrestrial ecosystems are found to lose carbon in the form of ecosystem respiration proportionally. Induced warming patterns and water limitation due to less precipitation are found to be the most dominant and significant sources driving negative trend of NBP from 1990s to 2000s.

It is important to note that the uncertainties in magnitudes of NEP and NBP in different studies are large (Patra et al., 2013) reflecting the complexity of the system and the sensitivity to the processes involved. Interactions amongst different environmental and anthropogenic forcings and their effect on ecosystems are nonlinear as reflected in our runs. For instance, impact of increasing $[CO_2]$ with changing LULCC is different from summing the impacts of changing $[CO_2]$ and LULCC separately. This nonlinear effect adds to the uncertainty in the estimated carbon fluxes. These uncertainties and disagreements between different studies point toward the need of more extensive research in this area focusing on individual environmental and anthropogenic factors and their impacts at finer spatial resolutions.

To contextualize our estimates of carbon fluxes, it is essential to include other major sources of carbon emission. Figure S7 depicts the carbon budget for India with fluxes of NEP, LULCC (E_luc), NBP, fire (E_fire) (van der Werf et al., 2010, but with updated burned area from Giglio et al., 2013), and fossil fuel (FF) emissions (Le Quéré et al., 2015) averaged for 2000–2012. These estimates show that the large magnitude of fossil fuel emissions clearly makes India a carbon source despite the fact that its terrestrial natural ecosystems are currently acting as a net sink for $[CO_2]$. Such comprehensive analysis is essential to develop more effective policies for environmental management.

It is to note that while net terrestrial CO₂ sink on a global scale has been reported to increase in recent years by global scale studies (Le Quéré et al., 2015), our study has shown opposite trend for India, mainly driven by increasing temperatures and decreasing precipitation. While greening in high latitudes has been reported due to warmer temperatures that have led to an increase in terrestrial CO₂ sink for these places, for a tropical country like India, increase in respiration has reversed the trend on a country scale. It can be inferred that with the current trend of increasing temperatures, tropical countries might be on a higher risk of losing carbon from their terrestrial ecosystems.

Most of the existing studies on global or regional scales have used multiple approaches to estimate the carbon fluxes on regional and global scales. This study, however, goes beyond flux estimation and focuses on understanding the sensitivity of net carbon fluxes for India to various environmental and anthropogenic factors, such as elevated CO_2 levels in the atmosphere, climate change, land cover and land use change, and nitrogen deposition. This is necessary to improve our understanding of processes responsible for sources and sink of carbon in terrestrial ecosystems. Country-level estimates are particularly relevant, for example, in the context of the 2015 Paris Climate agreement (UNFCCC, 2015), which requires quantifiable biosphere sources and sinks of carbon dioxide (CO_2) and other greenhouse gases at country level in order to enable successful implementation of climate policy.

Acknowledgments

The four sets of gridded terrestrial carbon fluxes outputs (GPP, NPP, NBP, Ra, Rh, and land use emission) simulated using ISAM for India under four different experimental discussed in the paper can be downloaded from the ISAM model website: http://climate.atmos.uiuc.edu/ Gahlot_etal_GRL_Data.tar.gz. These runs were conducted from 1801 to 2012, and here we provide the results from 1980 to 2012, which is the period the paper mainly focusing on. Part of the research was carried out under a visiting fellowship offered to Shilpa Gahlot by the University of Illinois under NASA grant (NNX414AD94G). Atul K Jain and Shijie Shu were supported by the NSF (NSF AGS 12-43071) and the US DOE (DOE DE-SC0016323). We thank Prabir Patra from JAMSTEC Japan for providing us the top-down modelestimated NBP results presented in Table 3.

References

- Ahlström, A., Schurgers, G., Arneth, A., & Smith, B. (2012). Robustness and uncertainty in terrestrial ecosystem carbon response to CMIP5 climate change projections. *Environmental Research Letters*, 7(4), 044008. https://doi.org/10.1088/1748-9326/7/4/044008
- Bala, G., Joshi, J., Chaturvedi, R. K., Gangamani, H. V., Hashimoto, H., & Nemani, R. (2013). Trends and variability of AVHRR-derived NPP in India. *Remote Sensing*, 5(2), 810–829. https://doi.org/10.3390/rs5020810
- Banger, K., Tian, H., Tao, B., Ren, W., Pan, S., Dangal, S., & Yang, J. (2015). Terrestrial net primary productivity in India during 1901–2010: Contributions from multiple environmental changes. *Climatic Change*, 132(4), 575–588. https://doi.org/10.1007/s10584-015-1448-5 Barman, R., Jain, A. K., & Liang, M. (2014a). Climate-driven uncertainties in modeling terrestrial gross primary production: A site level to
 - global-scale analysis. Global Change Biology, 20(5), 1394–1411. https://doi.org/10.1111/gcb.12474
- Barman, R., Jain, A. K., & Liang, M. (2014b). Climate-driven uncertainties in modeling terrestrial energy and water fluxes: A site-level to globalscale analysis. *Global Change Biology*, 20(6), 1885–1900. https://doi.org/10.1111/gcb.12473
- Beer, C., Reichstein, M., Tomelleri, E., Ciais, P., Jung, M., Carvalhais, N., ... Bondeau, A. (2010). Terrestrial gross carbon dioxide uptake: Global distribution and covariation with climate. Science, 329(5993), 834–838. https://doi.org/10.1126/science.1184984
- Cervarich, M., Shu, S., Jain, A. K., Arneth, A., Canadell, J., Friedlingstein, P., ... Zeng, N. (2016). The terrestrial carbon budget of South and Southeast Asia. *Environmental Research Letters*, *11*(10). https://doi.org/10.1088/1748-9326/11/10/105006
- Clark, D. B., Mercado, L. M., Sitch, S., Jones, C. D., Gedney, N., Best, M. J., ... Cox, P. M. (2011). The Joint UK Land Environment Simulator (JULES), model description—Part 2: Carbon fluxes and vegetation dynamics. *Geoscientific Model Development*, 4(3), 701–722. https://doi.org/ 10.5194/gmd-4-701-2011
- Dadhwal, V. K., Kushwaha, S. P. S., Singh, S., Patel, N. R., Nayak, R. K., Patil, P., ... Pujar, G. S. (2011). Recent results from EO studies on Indian carbon cycle assessment, ISPRS-International Archives of the Photogrammetry. *Remote Sensing and Spatial Information Sciences*, 3820, 3–9.
- El-Masri, B., Barman, R., Meiyappan, P., Song, Y., Liang, M., & Jain, A. K. (2013). Carbon dynamics in the Amazonian Basin: Integration of eddy covariance and ecophysiological data with a land surface model. *Agricultural and Forest Meteorology*, 182-183, 156–167. https://doi.org/ 10.1016/j.agrformet.2013.03.011
- El-Masri, B., Shu, S., & Jain, A. K. (2015). Implementation of a dynamic rooting depth and phenology into a land surface model: Evaluation of carbon, water, and energy fluxes in the high latitude ecosystems. *Agricultural and Forest Meteorology*, 211-212, 85–99. https://doi.org/ 10.1016/j.agrformet.2015.06.002
- FAO (2008). FAOSTAT. Food and Agriculture Organization of the United Nations, Rome, Italy. Retrieved from http://www.fao.org
- Ganshin, A., Oda, T., Saito, M., Maksyutov, S., Valsala, V., Andres, R. J., ... Nisbet, E. G. (2011). A global coupled Eulerian-Lagrangian model and 1× 1 km CO. *Geoscientific Model Development Discussion*, 4(3), 2047–2080. https://doi.org/10.5194/gmdd-4-2047-2011
- Giglio, L., Randerson, J. T., & van der Werf, G. R. (2013). Analysis of daily, monthly, and annual burned area using the fourth-generation global fire emissions database (GFED4). *Journal of Geophysical Research: Biogeosciences*, 118, 317–328. https://doi.org/10.1002/jgrg.20042
- Goroshi, S. K., Singh, R. P., Pradhan, R., & Parihar, J. S. (2014). Assessment of net primary productivity over India using Indian geostationary satellite (INSAT-3A) data, The International Archives of Photogrammetry. *Remote Sensing and Spatial Information Sciences*, XL-8(8), 561–568. https://doi.org/10.5194/isprsarchives-XL-8-561-2014
- Harris, I., Jones, P. D., Osborn, T. J., & Lister, D. H. (2014). Updated high-resolution grids of monthly climatic observations—The CRU TS3.10 dataset. *International Journal of Climatology*, 34(3), 623–642. https://doi.org/10.1002/joc.3711

- Ito, A., & Inatomi, M. (2012). Use of a process-based model for assessing the methane budgets of global terrestrial ecosystems and evaluation of uncertainty. *Biogeosciences*, 9(2), 759–773. https://doi.org/10.5194/bg-9-759-2012
- Jain, A. K., Meiyappan, P., Song, Y., & House, J. I. (2013). CO₂ emissions from land-use change affected more by nitrogen cycle, than by the choice of land-cover data. *Global Change Biology*, *19*(9), 2893–2906. https://doi.org/10.1111/gcb.12207
- Jain, A. K., Yang, X., Kheshgi, H., McGuire, A. D., Post, W. P., & Kicklighter, D. (2009). Nitrogen attenuation of terrestrial carbon cycle response to global environmental factors. *Global Biogeochemical Cycles*, 23, GB4028. https://doi.org/10.1029/2009GB003519
- Jung, M., Reichstein, M., & Bondeau, A. (2009). Towards global empirical upscaling of FLUXNET eddy covariance observations: Validation of a model tree ensemble approach using a biosphere model. *Biogeosciences*, 6(10), 2001–2013. https://doi.org/10.5194/bg-6-2001-2009
- Kaul, M., Dadhwal, V. K., & Mohren, G. M. J. (2009). Land use change and net C flux in Indian forests. Forest Ecology and Management, 258(2), 100–108. https://doi.org/10.1016/j.foreco.2009.03.049
- Klein, G. K., Beusen, A., van Drecht, G., & de Vos, M. (2011). The HYDE 3.1 spatially explicit database of human-induced land-use change over the past 12,000 years. *Global Ecology and Biogeography*, 20(1), 73–86. https://doi.org/10.1111/j.1466-8238.2010.00587.x
- Krinner, G., Viovy, N., de Noblet-Ducoudré, N., Ogée, J., Polcher, J., Friedlingstein, P., ... Prentice, I. C. (2005). A dynamic global vegetation model for studies of the coupled atmosphere-biosphere system. *Global Biogeochemical Cycles*, 19, GB1015. https://doi.org/10.1029/ 2003GB002199
- Lamarque, J.-F., Kyle, G. P., Meinshausen, M., Riahi, K., Smith, S. J., van Vuuren, D. P., ... Vitt, F. (2011). Global and regional evolution of shortlived radiatively-active gases and aerosols in the Representative Concentration Pathways. *Climatic Change*, 109(1–2), 191. https://doi.org/ 10.1007/s10584-011-0155-0
- Le Quéré, C., Moriarty, R., Andrew, R. M., Canadell, J. G., Sitch, S., Korsbakken, J. I., ... Houghton, R. A. (2015). Global carbon budget 2015. Earth System Science Data, 7(2), 349–396. https://doi.org/10.5194/essd-7-349-2015
- Meiyappan, P., & Jain, A. K. (2012). Three distinct global estimates of historical land-cover change and land-use conversions for over 200 years. *Frontiers of Earth Science*, 6(2), 122–139. https://doi.org/10.1007/s11707-012-0314-2
- Nayak, R. K., Mishra, N., & Dadhwal, V. K. (2016). Assessing the consistency between AVHRR and MODIS NDVI datasets for estimating terrestrial net primary productivity over India. Journal of Earth System Science, 125(6), 1189–1204. https://doi.org/10.1007/s12040-016-0723-9
- Nayak, R. K., Patel, N. R., & Dadhwal, V. K. (2010). Estimation and analysis of terrestrial net primary productivity over India by remote-sensingdriven terrestrial biosphere model. *Environmental Monitoring and Assessment*, 170(1-4), 195–213. https://doi.org/10.1007/s10661-009-1226-9
- Nayak, R. K., Patel, N. R., & Dadhwal, V. K. (2015). Spatio-temporal variability of net ecosystem productivity over India and its relationship to climatic variables. *Environmental Earth Sciences*, 74(2), 1743–1753. https://doi.org/10.1007/s12665-015-4182-4
- Oleson, K., Lawrence, D. M., Bonan, G. B., Drewniak, B., Huang, M., Koven, C. D., ... Yang, Z.-L. (2013). Technical description of version 4.5 of the Community Land Model (CLM) (NCAR Technical Note NCAR/TN-503+STR, pp. 420). https://doi.org/10.5065/D6RR1W7M
- Patra, P. K., Canadell, J. G., Houghton, R. A., Piao, S. L., Oh, N.-H., Ciais, P., ... Lasco, R. (2013). The carbon budget of South Asia. *Biogeosciences*, 10(1), 513–527. https://doi.org/10.5194/bg-10-513-2013
- Patra, P. K., Canadell, J. G., Thompson, R. L., Kondo, M., & Poulter, B. (2016). Sources and sinks of carbon dioxide in populous Asia, APN Project Reference: ARCP2013-01CMY-Patra/Canadell/. Retrieved from APN E-Lib: http://www.apn-gcr.org/resources
- Patra, P. K., Niwa, Y., Schuck, T. J., Brenninkmeijer, C. A. M., Machida, T., Matsueda, H., & Sawa, Y. (2011). Carbon balance of South Asia constrained by passenger aircraft CO₂ measurements. *Atmospheric Chemistry and Physics*, 11(9), 4163–4175. https://doi.org/10.5194/acp-11-4163-2011
- Peters, W., Jacobson, A. R., Sweeney, C., Andrews, A. E., Conway, T. J., Masarie, K., ... Worthy, D. E. (2007). An atmospheric perspective on North American carbon dioxide exchange: CarbonTracker. Proceedings of the National Academy of Sciences of the United States of America, 104(48), 18,925–18,930. https://doi.org/10.1073/pnas.0708986104
- Ramankutty, N., & Foley, J. (1999). Estimating historical changes in global land cover: Croplands from 1700 to 1992. Global Biogeochemical Cycles, 13(4), 997–1027. https://doi.org/10.1029/1999GB900046
- Rayner, P. J., Law, R. M., Allison, C. E., Francey, R. J., Trudinger, C. M., & Pickett-Heaps, C. (2008). Interannual variability of the global carbon cycle (1992–2005) inferred by inversion of atmospheric CO₂ and δ¹³CO₂ measurements. *Global Biogeochemical Cycles*, 22, GB3008. https:// doi.org/10.1029/2007GB003068
- Sasaki, T., Maki, T., Oohashi, S., & Akagi, K. (2003). Optimal sampling network and availability of data acquired at inland sites, Global Atmosphere Watch Report Series No. 148, World Meteorological Organization Global Atmosphere Watch (pp. 77–9).
- Sitch, S., Friedlingstein, P., Gruber, N., Jones, S. D., Murray-Tortarolo, G., Ahlström, A., ... Myneni, R. (2015). Recent trends and drivers of regional sources and sinks of carbon dioxide. *Biogeosciences*, *12*(3), 653–679. https://doi.org/10.5194/bg-12-653-2015
- Sitch, S., Smith, B., Prentice, I. C., Arneth, A., Bondeau, A., Cramer, W., ... Venevsky, S. (2003). Evaluation of ecosystem dynamics, plant geography and terrestrial carbon cycling in the LPJ dynamic global vegetation model. *Global Change Biology*, *9*(2), 161–185. https://doi.org/ 10.1046/j.1365-2486.2003.00569.x
- Song, Y., Cervarich, M., Jain, A. K., Kheshgi, H. S., William, L., & Cai, X. (2016). The interplay between bioenergy grass production and water resources in the United States. *Environmental Science & Technology*, *50*(6), 3010–3019. https://doi.org/10.1021/acs.est.5b05239
- Song, Y., Jain, A. K., & McIsaac, G. F. (2013). Implementation of dynamic crop growth processes into a land surface model: Evaluation of energy, water and carbon fluxes under corn and soybean rotation. *Biogeosciences*, 10(12), 8039–8066. https://doi.org/10.5194/bg-10-8039-2013
- Stocker, B. D., Feissli, F., Strassmann, K. M., Spahni, R., & Joos, F. (2014). Past and future carbon fluxes from land use change, shifting cultivation and wood harvest. *Tellus B*, 66(1), 23,188. https://doi.org/10.3402/tellusb.v66.23188
- Thompson, R. L., Patra, P. K., Chevallier, F., Maksyutov, S., Law, R. M., Ziehn, T., ... Ciais, P. (2016). Top-down assessment of the Asian carbon budget since the mid 1990s. *Nature Communications*, *7*, 10724. https://doi.org/10.1038/ncomms10724
- UNFCCC (2015). Adaptation of the Paris agreement, FCCC/CP/2015/L.9/Rev.1, 12 December 2015. Retrieved from https://unfccc.int/resource/ docs/2015/cop21/eng/l09r01.pdf
- van der Werf, G. R., Randerson, J. T., Giglio, L., Collatz, G. J., Mu, M., Kasibhatla, P. S., ... van Leeuwen, T. T. (2010). Global fire emissions and the contribution of deforestation, savanna, forest, agricultural, and peat fires (1997–2009). Atmospheric Chemistry and Physics, 10(23), 11.707–11.735.
- Zeng, N., Mariotti, A., & Wetzel, P. (2005). Terrestrial mechanisms of interannual CO₂ variability. *Global Biogeochemical Cycles*, *19*, GB1016. https://doi.org/10.1029/2004GB002273
- Zhao, M., & Running, S. W. (2010). Drought-induced reduction in global terrestrial net primary production from 2000 through 2009. Science, 329(5994), 940–943. https://doi.org/10.1126/science.1192666